\documentclass[twocolumn,twoside]{article}
\makeatletter\if@twocolumn\PassOptionsToPackage{switch}{lineno}\else\fi\makeatother

\usepackage{amsfonts,amssymb,amsbsy,latexsym,amsmath,tabulary,graphicx,times,xcolor}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% Following additional macros are required to function some 
% functions which are not available in the class used.
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\usepackage{url,multirow,morefloats,floatflt,cancel,tfrupee}
\makeatletter


\AtBeginDocument{\@ifpackageloaded{textcomp}{}{\usepackage{textcomp}}}
\makeatother
\usepackage{colortbl}
\usepackage{xcolor}
\usepackage{pifont}
\usepackage[nointegrals]{wasysym}
\urlstyle{rm}
\makeatletter

%%%For Table column width calculation.
\def\mcWidth#1{\csname TY@F#1\endcsname+\tabcolsep}

%%Hacking center and right align for table
\def\cAlignHack{\rightskip\@flushglue\leftskip\@flushglue\parindent\z@\parfillskip\z@skip}
\def\rAlignHack{\rightskip\z@skip\leftskip\@flushglue \parindent\z@\parfillskip\z@skip}

%Etal definition in references
\@ifundefined{etal}{\def\etal{\textit{et~al}}}{}


%\if@twocolumn\usepackage{dblfloatfix}\fi
\usepackage{ifxetex}
\ifxetex\else\if@twocolumn\@ifpackageloaded{stfloats}{}{\usepackage{dblfloatfix}}\fi\fi

\AtBeginDocument{
\expandafter\ifx\csname eqalign\endcsname\relax
\def\eqalign#1{\null\vcenter{\def\\{\cr}\openup\jot\m@th
  \ialign{\strut$\displaystyle{##}$\hfil&$\displaystyle{{}##}$\hfil
      \crcr#1\crcr}}\,}
\fi
}

%For fixing hardfail when unicode letters appear inside table with endfloat
\AtBeginDocument{%
  \@ifpackageloaded{endfloat}%
   {\renewcommand\efloat@iwrite[1]{\immediate\expandafter\protected@write\csname efloat@post#1\endcsname{}}}{\newif\ifefloat@tables}%
}%

\def\BreakURLText#1{\@tfor\brk@tempa:=#1\do{\brk@tempa\hskip0pt}}
\let\lt=<
\let\gt=>
\def\processVert{\ifmmode|\else\textbar\fi}
\let\processvert\processVert

\@ifundefined{subparagraph}{
\def\subparagraph{\@startsection{paragraph}{5}{2\parindent}{0ex plus 0.1ex minus 0.1ex}%
{0ex}{\normalfont\small\itshape}}%
}{}

% These are now gobbled, so won't appear in the PDF.
\newcommand\role[1]{\unskip}
\newcommand\aucollab[1]{\unskip}
  
\@ifundefined{tsGraphicsScaleX}{\gdef\tsGraphicsScaleX{1}}{}
\@ifundefined{tsGraphicsScaleY}{\gdef\tsGraphicsScaleY{.9}}{}
% To automatically resize figures to fit inside the text area
\def\checkGraphicsWidth{\ifdim\Gin@nat@width>\linewidth
	\tsGraphicsScaleX\linewidth\else\Gin@nat@width\fi}

\def\checkGraphicsHeight{\ifdim\Gin@nat@height>.9\textheight
	\tsGraphicsScaleY\textheight\else\Gin@nat@height\fi}

\def\fixFloatSize#1{}%\@ifundefined{processdelayedfloats}{\setbox0=\hbox{\includegraphics{#1}}\ifnum\wd0<\columnwidth\relax\renewenvironment{figure*}{\begin{figure}}{\end{figure}}\fi}{}}
\let\ts@includegraphics\includegraphics

\def\inlinegraphic[#1]#2{{\edef\@tempa{#1}\edef\baseline@shift{\ifx\@tempa\@empty0\else#1\fi}\edef\tempZ{\the\numexpr(\numexpr(\baseline@shift*\f@size/100))}\protect\raisebox{\tempZ pt}{\ts@includegraphics{#2}}}}

%\renewcommand{\includegraphics}[1]{\ts@includegraphics[width=\checkGraphicsWidth]{#1}}
\AtBeginDocument{\def\includegraphics{\@ifnextchar[{\ts@includegraphics}{\ts@includegraphics[width=\checkGraphicsWidth,height=\checkGraphicsHeight,keepaspectratio]}}}

\DeclareMathAlphabet{\mathpzc}{OT1}{pzc}{m}{it}

\def\URL#1#2{\@ifundefined{href}{#2}{\href{#1}{#2}}}

%%For url break
\def\UrlOrds{\do\*\do\-\do\~\do\'\do\"\do\-}%
\g@addto@macro{\UrlBreaks}{\UrlOrds}



\edef\fntEncoding{\f@encoding}
\def\EUoneEnc{EU1}
\makeatother
\def\floatpagefraction{0.8} 
\def\dblfloatpagefraction{0.8}
\def\style#1#2{#2}
\def\xxxguillemotleft{\fontencoding{T1}\selectfont\guillemotleft}
\def\xxxguillemotright{\fontencoding{T1}\selectfont\guillemotright}

\newif\ifmultipleabstract\multipleabstractfalse%
\newenvironment{typesetAbstractGroup}{}{}%

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\usepackage[authoryear]{natbib}



\newif\ifnokeywords\nokeywordsfalse

\makeatletter\input{size10-pointfive.clo}\makeatother%
\definecolor{kwdboxcolor}{RGB}{242,242,242}
\usepackage[hidelinks,colorlinks=true,allcolors=blue]{hyperref}
\linespread{1}                 
\def\floatpagefraction{0.8}

\usepackage[paperheight=11.69in,paperwidth=8.26in,top=1in,bottom=1in,left=1in,right=.75in,headsep=24pt]{geometry}

\usepackage{multirow-custom}

\makeatletter

\def\hlinewd#1{%
  \noalign{\ifnum0=`}\fi\hrule \@height #1%
  \futurelet\reserved@a\@xhline}
\def\tbltoprule{\hlinewd{1pt}\\[-14pt]}
\def\tblbottomrule{\noalign{\vspace*{6pt}}\hline\noalign{\vspace*{2pt}}}
\def\tblmidrule{\hline\noalign{\vspace*{2pt}}}



\let\@articleType\@empty
\let\@journalDoi\@empty
\let\@journalVolume\@empty
\let\@journalIssue\@empty
\let\@crossMarkLink\@empty
\let\@receivedDate\@empty
\let\@acceptedDate\@empty
\let\@revisedDate\@empty
\let\@publishedDate\@empty
\let\@copyrightYear\@empty
\let\@firstPage\@empty
\def\articleType#1{\gdef\@articleType{#1}}
\def\journalDoi#1{\gdef\@journalDoi{#1}}
\def\crossMarkLink#1{\gdef\@crossMarkLink{#1}}
\def\receivedDate#1{\gdef\@receivedDate{#1}}
\def\acceptedDate#1{\gdef\@acceptedDate{#1}}
\def\revisedDate#1{\gdef\@revisedDate{#1}}
\def\publishedDate#1{\gdef\@publishedDate{#1}}
\def\copyrightYear#1{\gdef\@copyrightYear{#1}}
\def\journalVolume#1{\gdef\@journalVolume{#1}}
\def\journalIssue#1{\gdef\@journalIssue{#1}}
\def\firstPage#1{\gdef\@firstPage{#1}}
\let\@issuelogo\@empty%
\let\@issuelogoex\@empty%
\def\issuelogo#1{\gdef\@issuelogo{#1}}
\def\issuelogoex#1{\gdef\@issuelogoex{#1}}


\def\author#1{%
  \gdef\@author{%
    \hskip-\dimexpr(\tabcolsep)\hskip5pt%
    \parbox{\dimexpr\textwidth-1pt}%
    {\fontsize{11}{13}\selectfont\raggedright #1}%
  }%
}


\usepackage{pharmascope-abs}
\usepackage{caption}
\usepackage{lastpage}
\usepackage{fancyhdr}
\usepackage[noindentafter,explicit]{titlesec}
\usepackage{fontspec}

\setmainfont[%
BoldFont=cambriab.ttf,%
ItalicFont=CAMBRIAI.TTF,%
BoldItalicFont=CAMBRIAZ.TTF]{Cambria.ttf}
\lefthyphenmin = 3

\def\title#1{%
  \gdef\@title{%
      \vspace*{-40pt}%
       \ifx\@articleType\@empty\else{\fontsize{10}{12}\scshape\selectfont\hspace{8pt}\@articleType\hfill\mbox{}\par\vspace{2pt}}\fi%
       \minipage{\linewidth}
         \hrulefill\\[-0.7pt]%%{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}\hfill
         \mbox{~}\hspace{5pt}\parbox{.1\linewidth}{\IfFileExists{images/\@issuelogo.\@issuelogoex}{\includegraphics[width=79pt,height=61pt]{images/\@issuelogo.\@issuelogoex}}{\includegraphics[width=75pt,height=50pt]{ijrps_logo.png}}}\hfill
         \fcolorbox{kwdboxcolor}{kwdboxcolor}{\parbox{.792\linewidth}{%
          \begin{center}\fontsize{17}{17}\selectfont\scshape\vskip-7pt International Journal of Research in Pharmaceutical Sciences\hfill\end{center}%
          \vspace*{-10pt}\hspace*{4pt}{\fontsize{8}{9}\selectfont Published by JK Welfare \& Pharmascope Foundation\hfill  Journal Home Page: \href{www.ijrps.com}{\color{blue}\underline{\smash{www.ijrps.com}}}}\hspace*{4pt}\mbox{}}}%
       \par\vspace*{-1pt}\rule{\linewidth}{1.3pt}%
       \endminipage%
    \par\vspace*{9.2pt}\parbox{.98\linewidth}{\linespread{.9}\raggedright\fontsize{14}{17}\selectfont #1}%
    \vspace*{-8pt}%
  }
}


\setlength{\parindent}{0pt}
\setlength{\parskip}{0.4pc plus 1pt minus 1pt}

\def\abbrvJournalTitle{Int. J. Res. Pharm. Sci.}

\fancypagestyle{headings}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
}\pagestyle{headings}

                                                         
\fancypagestyle{plain}{%
  \renewcommand{\headrulewidth}{0pt}%
  \renewcommand{\footrulewidth}{0.3pt}%
  \fancyhf{}%
  \fancyhead[R]{%
    \fontsize{9.12}{11}\selectfont\RunningAuthor,\ \abbrvJournalTitle,\ \ifx\@journalVolume\@empty X\else\@journalVolume\fi%
    \ifx\@journalIssue\@empty\else(\@journalIssue)\fi%
    ,\ \ifx\@firstPage\@empty 1\else\@firstPage\fi-\pageref*{LastPage}%
  }%
  \fancyfoot[LO,RE]{\fontsize{9.12}{11}\selectfont\textcopyright\ International Journal of Research in Pharmaceutical Sciences}%
  \fancyfoot[RO,LE]{\fontsize{9.12}{11}\selectfont\thepage}
  \ifx\@firstPage\@empty\else\setcounter{page}{\@firstPage}\fi
}


\def\NormalBaseline{\def\baselinestretch{1.1}}
\usepackage{textcase}
\setcounter{secnumdepth}{0}

\titleformat{\section}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{13}\selectfont}
{\thesection}
{6pt}
{\MakeTextUppercase{#1}}
[]
\titleformat{\subsection}[block]{\bfseries\boldmath\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsection}
{6pt}
{#1}
[]
\titleformat{\subsubsection}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubsubsection}
{6pt}
{#1}
[]
\titleformat{\paragraph}[block]{\NormalBaseline\filright\fontsize{10.5}{10}\selectfont}
{\theparagraph}
{6pt}
{#1}
[]
\titleformat{\subparagraph}[block]{\NormalBaseline\filright\fontsize{10.5}{12}\selectfont}
{\thesubparagraph}
{6pt}
{#1}
[]

\titlespacing{\section}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subsubsection}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\paragraph}{0pt}{.5\baselineskip}{.5\baselineskip}  
\titlespacing{\subparagraph}{0pt}{.5\baselineskip}{.5\baselineskip}

\captionsetup[figure]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}
\captionsetup[table]{skip=1.4pt,font=bf,labelsep=colon,justification=raggedright,singlelinecheck=false}

\def\bibyear#1{#1}
%\def\bibjtitle#1{#1}   %%Publisher request
\def\bibauand{}
\setlength\bibsep{3pt}
\setlength\bibhang{8pt}

\makeatother
\date{}


\usepackage{float}

\begin{document}


\def\RunningAuthor{Koya Prabhakara Rao et al.}
\firstPage{2210}
\articleType{Original Article}
\receivedDate{20 Feb 2020}
\acceptedDate{04 Apr 2020}
\revisedDate{10 Mar 2020}
\journalVolume{2020, 11}
\journalIssue{2}
\journalDoi{ijrps.v11i2.2172}
\copyrightYear{2020}
\def\authorCount{5}
\def\affCount{3}

\def\journalTitle{International Journal of Research in Pharmaceutical Sciences}

\title{Bioanalytical LC-MS/MS method for Determination and comparison of Selexipag Assay in various Biological materials and its Application to Pharmacokinetics Studies in Rat plasma}
\author{Namburi~LA~Amara~babu\textsuperscript{1},  
            Kalyani~Koganti\textsuperscript{1},  
            Babji~Palakeeti\textsuperscript{2},  
            Koduri~SV~Srinivas\textsuperscript{3}, 
            Koya~Prabhakara~Rao\textsuperscript{*}\textsuperscript{1}~\\[5pt]{\textsuperscript{1}Department of Science and Humanities, New Generation Materials Lab (NGML), Vignan's Foundation for Science Technology and Research University (VFSTR) (Deemed to be University), Vadlamudi, Guntur-522213, Andhra Pradesh, India}~\\{\textsuperscript{2}Department of Chemistry, National Institute of Technology Warangal, Warangal-506004, Telangana, India}~\\{\textsuperscript{3}Syngene International LTD, Bommasandra, Bangalore-560099, Karnataka, India}}

\begin{abstract}
A rapid, sensitive and selective bioanalytical method was developed and validated by Liquid Chromatography - Mass spectrometry (LC-MS/MS) for determination and comparison of Selexipag\% assay in various biological materials. Selexipag was extracted and compared its \% assay after protein precipitation technique from various biological materials such as rat plasma, rabbit plasma, human plasma and urine. Ambrisentan was selected as internal standard. Selected analytical column Waters, X-Bridge C18 3.5\ensuremath{\mu} (150 x 4.6 mm), mobile phase consists of Hexane sulfonic acid and Acetonitrile (80:20 v/v) at a flow rate of 1.0 mL /min in isocratic mode and Selexipag was determined by the +ve mode of electrospray ionization by using Mass detector. The method was developed to assess the lower limit of detection (LLOD)(0.5 ng/mL), lower limit of quantification(LLOQ) (5 ng/mL) concentrations and Linearity range of 1 ng/mL to 20 ng/mL concentration with regression correlation coefficient 0.999 were observed for Selexipag in Rat plasma. The test samples at lower, medium and higher concentrations of Selexipag shows precision (\% CV was 0.8 to 1.11) and accuracy results (97.3 \% to 100.6\%) for inter-day and intra-day analysis at 1, 5, 10, 15 ng/mL concentrations of Selexipag. Stability of Selexipag exists in all conditions like wet extract, bench top, freeze-thaw and in instrument auto sampler as per FDA guidelines. 
\end{abstract}\def\keywordstitle{Keywords}

\begin{keywords}Selexipag,\newline Rat Plasma,\newline LC-MS/MS,\newline Pharmacokinetics,\newline Bioanalytical methods
\end{keywords}
\twocolumn[ \maketitle {\printKwdAbsBox}]
\makeatletter{\raggedright\textsuperscript{*}Corresponding Author\par Name:\ Koya~Prabhakara~Rao~\\ Phone:\ +91-967617858~\\  Email:\ kprao2005@gmail.com
        \par}\vspace*{-11pt}\hrulefill\par{\fontsize{12}{14}\selectfont ISSN: 0975-7538}\par% 
        \textsc{DOI:}\ \href{https://doi.org/10.26452/\@journalDoi}{\textcolor{blue}{\underline{\smash{https://doi.org/10.26452/\@journalDoi}}}}\par%
        \vspace*{-11pt}\hrulefill\\{\fontsize{9.12}{10.12}\selectfont Production and Hosted by}\par{IJRPS |~\href{www.ijrps.com}{www.ijrps.com}}\par%
        \vspace*{-2pt}{\fontsize{9.12}{10.12}\selectfont\textcopyright\ \@copyrightYear\ $|$ All rights reserved.}\par%
        \vspace*{-11pt}\rule{\linewidth}{1.2pt}
        \makeatother
        
    
\section{Introduction}
Pulmonary arterial hypertension (PAH) is one the dangerous disease caused by high pulmonary arterial pressure in the presence of a normal or reduced cardiac output \unskip~\citep{739766:17569135}.  It has known that, for the treatment of PAH of prostacyclin analogues used due to dysregulation of prostacyclin pathways involved in the pathogenesis of PAH. In fact, the first targeted PAH therapy to be approved was a synthetic prostacyclin analogue, Epoprostenol, which was executed as a regular intravenous infusion method \unskip~\citep{739766:17569146,739766:17569139,739766:17569136}. However, this is a very complex execution process caused serious side-effects lead to limit its use for the treatment of PAH \unskip~\citep{739766:17569129}. Alternatives to Epoprostenol some other drugs were used in the USA including treprostinil and iloprost, which also have some weakness with respect to injection site pain (treprostinil), frequent dosing (iloprost), and typical prostanoid-related side-effects such as diarrhea, jaw pain, flushing and headache \unskip~\citep{739766:17569145,739766:17569143}. An oral prostacyclin treatment method would be a major favor for PAH treatment. However, most of the oral prostacyclin analogues, such as treprostinil and beraprost, have failed to display a stubborn treatment response as shown by the primary efficacy end-point \unskip~\citep{739766:17569141}.


\begin{table*}[!htbp]
\caption{\boldmath {Method development Trails} }
\label{tw-f59b203916f2}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.14620000000000005\linewidth-2\tabcolsep}p{\dimexpr.3538\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}p{\dimexpr.25\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Sl. No. & Mobile Phase (v/v) & Column & Observation\\
\tblmidrule 
Trial-1 &
  0.1\% OPA: ACN (20:80) &
  Zorbax C18 (250x4.6mm, 5.0\ensuremath{\mu}) &
  Peak retention time is very low in Figure~\ref{f-b6caea5a6b6f}\\
Trial-2 &
  0.1\% OPA:ACN (50:50) &
  Zorbax C18 (250x4.6mm, 5.0\ensuremath{\mu}) &
  Base line not sufficientin  \mbox{}\protect\newline Figure~\ref{f-61092186cb0c}\\
Trial-3 &
  0.1\% TEA pH-2.5 with OPA and ACN (20:80) &
  Waters X-Bridge C18 (150x4.6mm, 3.5\ensuremath{\mu}) &
  Peaks resolution is very lowin   Figure~\ref{f-63089a65903c}\\
Trial-4 &
  2.5gm Hexane Sulfonic acid in 1lt water adjusted pH-2.5  with OPA : ACN (50:50) &
  Waters X-Bridge C18 (150x4.6mm, 3.5\ensuremath{\mu}) &
  Peaks resolution is very lowin   Figure~\ref{f-c9e5e9b9e9fc}\\
Trial-5 &
  2.5gm Hexane Sulfonic acid in 1lt water adjusted pH-2.5 with OPA : ACN (80:20) &
  Waters X-Bridge C18 (150x4.6mm, 3.5\ensuremath{\mu}) &
  All the system suitability results are within the limit in Figure~\ref{f-ae40b2084f03}\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Matrix effects and extraction recoveries for Selexipagin rat plasma (n=6)} }
\label{tw-711742c4376d}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3438\linewidth-2\tabcolsep}p{\dimexpr.31619999999999997\linewidth-2\tabcolsep}p{\dimexpr.34\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Spiked concentration \mbox{}\protect\newline (ng/ml) & Matrix effect (\%) \mbox{}\protect\newline (mean \ensuremath{\pm} SD, n=6)  & Recovery (\%) \mbox{}\protect\newline (mean \ensuremath{\pm} SD, n=6)\\
\tblmidrule 
5.0 &
  97.85\ensuremath{\pm}5.05 &
  98.21\ensuremath{\pm}2.36\\
10.0 &
  99.66\ensuremath{\pm}10.15 &
  99.89\ensuremath{\pm}6.22\\
15.0 &
  98.63\ensuremath{\pm}8.38 &
  98.92\ensuremath{\pm}5.37\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Linearity data for Selexipag calibration standards} }
\label{tw-95998a124418}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.1682\linewidth-2\tabcolsep}p{\dimexpr.1986\linewidth-2\tabcolsep}p{\dimexpr.2055\linewidth-2\tabcolsep}p{\dimexpr.2277\linewidth-2\tabcolsep}p{\dimexpr.20\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Linearity & Selexipag Conc. (ng/ml) & Selexipag \mbox{}\protect\newline Peak response & Internal Standard (IS) Peak response & Area ratio\\
\tblmidrule 
Linearity-1 &
  1 &
  0.105 &
  1.021 &
  0.103\\
Linearity-2 &
  3 &
  0.254 &
  1.022 &
  0.249\\
Linearity-3 &
  5 &
  0.526 &
  1.026 &
  0.513\\
Linearity-4 &
  8 &
  0.754 &
  1.024 &
  0.736\\
Linearity-5 &
  10 &
  1.051 &
  1.027 &
  1.023\\
Linearity-6 &
  12.5 &
  1.259 &
  1.025 &
  1.228\\
Lineaity-7 &
  15 &
  1.529 &
  1.021 &
  1.498\\
Linearity-8 &
  20 &
  2.055 &
  1.022 &
  2.011\\
Slope &
  \multicolumn{4}{p{\dimexpr(.8317999999999999\linewidth-2\tabcolsep)}}{0.009980}\\
Intercept &
  \multicolumn{4}{p{\dimexpr(.8317999999999999\linewidth-2\tabcolsep)}}{0.00273}\\
Correlation coefficient &
  \multicolumn{4}{p{\dimexpr(.8317999999999999\linewidth-2\tabcolsep)}}{0.999}\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Inter-run and between-run precision and accuracy for Selexipag} }
\label{tw-c8dfd0913546}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.1667\linewidth-2\tabcolsep}p{\dimexpr.11329999999999998\linewidth-2\tabcolsep}p{\dimexpr.14\linewidth-2\tabcolsep}p{\dimexpr.1598\linewidth-2\tabcolsep}p{\dimexpr.1202\linewidth-2\tabcolsep}p{\dimexpr.16\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Nominal Conc. \mbox{}\protect\newline (ng/ml) & \multicolumn{3}{p{\dimexpr(.42\linewidth-2\tabcolsep)}}{\cAlignHack Inter run} & \multicolumn{3}{p{\dimexpr(.44\linewidth-2\tabcolsep)}}{\cAlignHack Between run}\\
\rowcolor{kwdboxcolor} & Average (mean \ensuremath{\pm} SD, n=6, ng/mL) & Precision \mbox{}\protect\newline (\%CV) & Accuracy (\%) & Average \mbox{}\protect\newline (mean \ensuremath{\pm} SD, n=6, ng/mL) & Precision \mbox{}\protect\newline (\%CV) & Accuracy (\%)\\
\tblmidrule 
 &
   &
   &
   &
   &
   &
  \\
1.0 &
  0.99 &
  1.05 &
  98.2 &
  5.02 &
  1.04 &
  97.3\\
5.0 &
  5.01 &
  1.11 &
  99.5 &
  49.98 &
  1.07 &
  99.2\\
10.0 &
  10.01 &
  1.01 &
  100.1 &
  100.11 &
  1.00 &
  100.05\\
15.0 &
  14.99 &
  0.87 &
  100.6 &
  150.04 &
  0.86 &
  100.17\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Stability of the Selexipag samples} }
\label{tw-2906e72760bc}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.3217\linewidth-2\tabcolsep}p{\dimexpr.1488\linewidth-2\tabcolsep}p{\dimexpr.20139999999999997\linewidth-2\tabcolsep}p{\dimexpr.2166\linewidth-2\tabcolsep}p{\dimexpr.11149999999999999\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}\multicolumn{2}{p{\dimexpr(.47050000000000004\linewidth-2\tabcolsep)}}{Stability experiments} & Spiked plasma concentration(mean \ensuremath{\pm} SD, n=6, ng/mL)  & Concentration measured (mean \ensuremath{\pm} SD, n=6, ng/mL) & \%CV \mbox{}\protect\newline (n=6)\\
\tblmidrule 
\multicolumn{1}{p{\dimexpr(.3217\linewidth-2\tabcolsep)}}{\multirow{2}{\linewidth}{Benchtop stability  \mbox{}\protect\newline (for 24h)}} &
  LQC &
  1.0 &
  0.98 &
  1.5\\
 &
  HQC &
  15.0 &
  15.02 &
  0.7\\
\multicolumn{1}{p{\dimexpr(.3217\linewidth-2\tabcolsep)}}{\multirow{2}{\linewidth}{Autosampler stability (6\ensuremath{^\circ}C for 24h)}} &
  LQC &
  1.0 &
  1.01 &
  1.2\\
 &
  HQC &
  15.0 &
  14.99 &
  1.5\\
\multicolumn{1}{p{\dimexpr(.3217\linewidth-2\tabcolsep)}}{\multirow{2}{\linewidth}{Long term stability \mbox{}\protect\newline (for 27-days)}} &
  LQC &
  1.0 &
  1.01 &
  0.6\\
 &
  HQC &
  15.0 &
  15.01 &
  1.2\\
\multicolumn{1}{p{\dimexpr(.3217\linewidth-2\tabcolsep)}}{\multirow{2}{\linewidth}{Freeze-thaw stability (for 24h)}} &
  LQC &
  1.0 &
  0.99 &
  0.8\\
 &
  HQC &
  15.0 &
  15.00 &
  1.1\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Mean pharmacokinetic parameters of Selexipag in rat plasma} }
\label{tw-3b347697bda3}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.49129999999999995\linewidth-2\tabcolsep}p{\dimexpr.50870000000000005\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Pharmacokinetic Parameters & Selexipag\\
\tblmidrule 
AUC0-t  (ng h/ml) &
  3\\
Cmax  (ng/ml) &
  12.5\\
AUC0-\ensuremath{\infty }  (ng h/ml) &
  17\\
Kel &
  0.5\\
t1/2 &
  1.38\\
tmax (h) &
  2.0\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\begin{table*}[!htbp]
\caption{\boldmath {Comparison of Selexipag \% recovery between rat plasma, rabbit plasma, human plasma and rat urine } }
\label{tw-891b4369db50}
\def\arraystretch{1.1}
\ignorespaces 
\centering 
\begin{tabulary}{\linewidth}{p{\dimexpr.283\linewidth-2\tabcolsep}p{\dimexpr.14729999999999999\linewidth-2\tabcolsep}p{\dimexpr.1987\linewidth-2\tabcolsep}p{\dimexpr.2332\linewidth-2\tabcolsep}p{\dimexpr.13780000000000001\linewidth-2\tabcolsep}}
\tbltoprule \rowcolor{kwdboxcolor}Selexipag Extraction in & Rat plasma & Rabbit plasma & Human plasma & Rat urine\\
\rowcolor{kwdboxcolor}Average recovery in \% &  &  &  & \\
\tblmidrule 
Six replicates of HQC &
  94.74 &
  59.71 &
  57.86 &
  80.37\\
Six replicates of MQC &
  97.12 &
  60.46 &
  61.36 &
  82.78\\
Six replicates of LQC &
  95.21 &
  68.83 &
  77.40 &
  95.78\\
Six replicates of LLQC &
  94.68 &
  55.40 &
  70.42 &
  91.16\\
\tblbottomrule 
\end{tabulary}\par 
\end{table*}

\bgroup
\fixFloatSize{images/328d2ee1-71dd-47c9-a6b3-194d594b9530-upicture1.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/328d2ee1-71dd-47c9-a6b3-194d594b9530-upicture1.png}{\includegraphics{images/328d2ee1-71dd-47c9-a6b3-194d594b9530-upicture1.png}}{}
\makeatother 
\caption{\boldmath {Mass spectrum of Selexipag (analyte)}}
\label{f-d82d864b5527}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/f4d25e56-c274-466e-ab76-44a0d7059214-upicture2.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/f4d25e56-c274-466e-ab76-44a0d7059214-upicture2.png}{\includegraphics{images/f4d25e56-c274-466e-ab76-44a0d7059214-upicture2.png}}{}
\makeatother 
\caption{\boldmath {Mass spectrum of Ambrisentan (internal standard)}}
\label{f-fd994b7b7d94}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2f2a978b-a106-42ff-8598-5e8ce1f8c7a7-upicture3.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2f2a978b-a106-42ff-8598-5e8ce1f8c7a7-upicture3.png}{\includegraphics{images/2f2a978b-a106-42ff-8598-5e8ce1f8c7a7-upicture3.png}}{}
\makeatother 
\caption{\boldmath {Chromatogram for Trial-1}}
\label{f-b6caea5a6b6f}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/4edd8c5c-8909-408f-a319-972f91ba7061-upicture4.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/4edd8c5c-8909-408f-a319-972f91ba7061-upicture4.png}{\includegraphics{images/4edd8c5c-8909-408f-a319-972f91ba7061-upicture4.png}}{}
\makeatother 
\caption{\boldmath {Chromatogramfor Trial-2}}
\label{f-61092186cb0c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/fbe2670e-1d9a-4589-b2e2-d9825a61c959-upicture5.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/fbe2670e-1d9a-4589-b2e2-d9825a61c959-upicture5.png}{\includegraphics{images/fbe2670e-1d9a-4589-b2e2-d9825a61c959-upicture5.png}}{}
\makeatother 
\caption{\boldmath {Chromatogram for Trial-3 }}
\label{f-63089a65903c}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/92336adf-64ee-4af6-987c-dc625ed05f12-upicture6.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/92336adf-64ee-4af6-987c-dc625ed05f12-upicture6.png}{\includegraphics{images/92336adf-64ee-4af6-987c-dc625ed05f12-upicture6.png}}{}
\makeatother 
\caption{\boldmath {Chromatogram for Trials-4}}
\label{f-c9e5e9b9e9fc}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/9ece3d48-949c-48fc-a4ce-966888c81ed2-upicture7.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/9ece3d48-949c-48fc-a4ce-966888c81ed2-upicture7.png}{\includegraphics{images/9ece3d48-949c-48fc-a4ce-966888c81ed2-upicture7.png}}{}
\makeatother 
\caption{\boldmath {Optimized method chromatogram}}
\label{f-ae40b2084f03}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/54f49f3a-abb7-49b7-b353-e7c5684d86c3-upicture8.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/54f49f3a-abb7-49b7-b353-e7c5684d86c3-upicture8.png}{\includegraphics{images/54f49f3a-abb7-49b7-b353-e7c5684d86c3-upicture8.png}}{}
\makeatother 
\caption{\boldmath {Solid phase extraction }}
\label{f-9d61fde07332}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/33dd8df5-5d32-40ed-8e63-51431ae860aa-upicture9.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/33dd8df5-5d32-40ed-8e63-51431ae860aa-upicture9.png}{\includegraphics{images/33dd8df5-5d32-40ed-8e63-51431ae860aa-upicture9.png}}{}
\makeatother 
\caption{\boldmath {Protein Precipitation Extraction}}
\label{f-401ed6a649df}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8ef90b52-19f4-4bca-9b78-4c6c106adc4a-upicture10.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8ef90b52-19f4-4bca-9b78-4c6c106adc4a-upicture10.png}{\includegraphics{images/8ef90b52-19f4-4bca-9b78-4c6c106adc4a-upicture10.png}}{}
\makeatother 
\caption{\boldmath {Blank plasma chromatogram for Selexipag in rat plasma}}
\label{f-de2e98ae52f5}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c092a755-8db4-49e9-ba62-92facb249cbe-upicture11.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c092a755-8db4-49e9-ba62-92facb249cbe-upicture11.png}{\includegraphics{images/c092a755-8db4-49e9-ba62-92facb249cbe-upicture11.png}}{}
\makeatother 
\caption{\boldmath {LLOQ QC chromatogram for Selexipag in rat plasma}}
\label{f-263f6722af10}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/e6f917a2-ca31-44d2-80cf-1aee3f5ce60e-upicture12.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/e6f917a2-ca31-44d2-80cf-1aee3f5ce60e-upicture12.png}{\includegraphics{images/e6f917a2-ca31-44d2-80cf-1aee3f5ce60e-upicture12.png}}{}
\makeatother 
\caption{\boldmath {Calibration plot for Concentration v/s Area ratio of Selexipag and Internal Standard}}
\label{f-353c714de4d4}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/dffb16b9-a622-40b7-9b12-f9be3f48c5ab-upicture13.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/dffb16b9-a622-40b7-9b12-f9be3f48c5ab-upicture13.png}{\includegraphics{images/dffb16b9-a622-40b7-9b12-f9be3f48c5ab-upicture13.png}}{}
\makeatother 
\caption{\boldmath {Recovery plot for Selexipag in Ratplasma (Concentration in ng/mL Vs Time in Hours)}}
\label{f-d93420f77a65}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/6c8077f3-28d9-46e1-a578-ec16f6f2d7a0-upicture14.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/6c8077f3-28d9-46e1-a578-ec16f6f2d7a0-upicture14.png}{\includegraphics{images/6c8077f3-28d9-46e1-a578-ec16f6f2d7a0-upicture14.png}}{}
\makeatother 
\caption{\boldmath {Blank plasma chromatogram for Selexipag in rabbit plasma}}
\label{f-c00fedce4e01}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/cd5f7805-a8ca-4e3f-b556-e6769562af32-upicture15.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/cd5f7805-a8ca-4e3f-b556-e6769562af32-upicture15.png}{\includegraphics{images/cd5f7805-a8ca-4e3f-b556-e6769562af32-upicture15.png}}{}
\makeatother 
\caption{\boldmath {Blank plasma chromatogram for Selexipag in human plasma}}
\label{f-9e9ee4dd42b6}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/8b40c92a-8ce8-4ca2-a035-8d6824cd7916-upicture16.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/8b40c92a-8ce8-4ca2-a035-8d6824cd7916-upicture16.png}{\includegraphics{images/8b40c92a-8ce8-4ca2-a035-8d6824cd7916-upicture16.png}}{}
\makeatother 
\caption{\boldmath {Blank plasma chromatogram for Selexipag in rat urine}}
\label{f-9c19dfe6177e}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/2cc6b7e3-f971-476b-b98d-29bd49314066-upicture17.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/2cc6b7e3-f971-476b-b98d-29bd49314066-upicture17.png}{\includegraphics{images/2cc6b7e3-f971-476b-b98d-29bd49314066-upicture17.png}}{}
\makeatother 
\caption{\boldmath {LLOQ QC chromatogram for Selexipag in rabbit plasma}}
\label{f-6d6c7bd37f26}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/c286c673-b9da-4679-a69e-e24cb2128876-upicture18.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/c286c673-b9da-4679-a69e-e24cb2128876-upicture18.png}{\includegraphics{images/c286c673-b9da-4679-a69e-e24cb2128876-upicture18.png}}{}
\makeatother 
\caption{\boldmath {LLOQ QC chromatogram for Selexipag in human plasma}}
\label{f-dd304ba25634}
\end{figure}
\egroup

\bgroup
\fixFloatSize{images/d1622566-8c2f-4c34-9b65-aa8881d171a6-upicture19.png}
\begin{figure}[!htbp]
\centering \makeatletter\IfFileExists{images/d1622566-8c2f-4c34-9b65-aa8881d171a6-upicture19.png}{\includegraphics{images/d1622566-8c2f-4c34-9b65-aa8881d171a6-upicture19.png}}{}
\makeatother 
\caption{\boldmath {LLOQ QC chromatogram for Selexipag in rat urine}}
\label{f-1de126a74b9f}
\end{figure}
\egroup
The drug substance Selexipag is developed by Actelion of oral treatment for selective IP receptor agonist. In fact, it can convert to an effective metabolite \unskip~\citep{739766:17569143} by rapidly hydrolyzed in the presence of hepatic microsomes. Moreover,  Selexipag (Figure~\ref{f-d82d864b5527}) and its active metabolite (ACT-333679, 37-fold potent than parent analyte) are agonists \unskip~\citep{739766:17569142}  have a greater binding affinity for the human Prostacyclin receptor (IP receptor) compared with any other prostanoid receptor \unskip~\citep{739766:17569130}  leads to vasodilation \unskip~\citep{739766:17569126} in the pulmonary circulation \unskip~\citep{739766:17569128}. Usage of  Selexipag along with strong inhibitors of the liver enzyme such as gemfibrozil \unskip~\citep{739766:17569138} contraindicated in Europe. It is because the concentrations of Selexipag increased to twofold, which leads to active metabolite increased to 11-fold caused likely more inimical effects \unskip~\citep{739766:17569139}. In fact, selexipag and its active metabolite are chemically different from prostacyclin with peculiar pharmacology have similar modes of action to that of internal prostacyclin. Therefore, selexipag may be an alternative fascinating oral prostacyclin analogue for the treatment of PAH. The current proof-of-concept study was created to assess the safety, tolerability and efficacy of selexipag in adult patients with emblematic PAH. In fact, the inimical effects of Selexipag are similar to those of intravenous prostacyclins \unskip~\citep{739766:17569137} used for the treatment of PAH. However, in the usage of Selexipag drug, commonly occurred side effects are headache and jaw pain. Moreover, in people taking Selexipag also suffered a high risk of increased hyperthyroidism \unskip~\citep{739766:17569131}. By applying, an accurate and precise validated method for the treatment of PAH using Selexipag avoided or reduced side effects. 

Ambrisentan was selected as internal standard, since this this drug can be used for PAH treatment (Figure~\ref{f-fd994b7b7d94}). Bioanalysis is an area of analytical chemistry \unskip~\citep{739766:17569144} in the quantitative measurement of xenobiotics \unskip~\citep{739766:17569132,739766:17569133,739766:17569140}  and biotics in biological systems. Bioanalytical common techniques reported in the literature are, hyphenated techniques like LC-MS, GC-MS, LC-DAD, CE-MS, chromatographic methods like HPLC, GC, UPLC, Supercritical fluid chromatography. Bioanalytical \unskip~\citep{739766:17569147} methods are generally deal with biological samples consists of the analyte along with a distinct range of chemicals that can have a conflicting impact on their accurate and precise quantification. As such, a wide range of techniques reported extracting the analyte from its matrix such as protein precipitation, solid phase extraction and liquid-liquid extraction.

In this context, we are interested in  developing simple, specific, accurate and precise various chromatographic and spectroscopic methods for validation of various drugs. For quantification of Selexipag drug one HPLC method \unskip~\citep{739766:17569144} and three spectrophotometric methods were reported in the literature. However, no LC-MS method reported so far to the best of our knowledge. In this study, we report on the development and validation of an LC-MS/MS method for quantification of Selexipag in Rat plasma. Moreover, we also report the application for this LC-MS/MS method to its pharmacokinetics studies.
    
\section{Materials and Methods}
\textbf{Chemicals and Standards}

The APIs of Selexipag (purity{\textgreater} 99\%) and Ambrisentan(purity {\textgreater} 99\%) were procured from Glenmark Pharmaceuticals, Mumbai. LCMS grade of acetonitrile was procured from J.T. Baker. HPLC grade of hexane sulfonic acid and orthophosphoric acid were procured from Merck and rat plasma was purchased from Bharat Biotech, Hyderabad.

\textbf{Instrumentation}

An instrument HPLC system (Waters Alliance e2695 model), connected with LCMS (QTRAP 5500 triple quadrupole) was used to develop the method and its validation. Empower 2.0 software was used for processing the data.

\textbf{Detection}

The LCMS instrument functioned in MRM (multiple reaction monitoring) modes and a +ve mode of electrospray ionization is used to sample introduction. The optimized conditions were as follows: the selected flow of nebulizer gas was 30 psi, the flow of curtain gas -25 psi, voltage of ion-spray 2000v and the temperature was 375 \ensuremath{^\circ}C. The potentials of declustering pressure (DP), focusing pressure (FP), entrance pressure (EP), collision energy (CE), cell exist (CXP) for Selexipag are 45, 35, 10, 8ev and 12 respectively. The CAD (Collision activated dissociation) gas was adjusted to 4psi with nitrogen gas. Both quadrupoles 1 and 3 were continued at a single resolution and the time was 300ms. The changes in mass were selected as m/z 462.5 to 496.2 for Selexipag and 378.5 to 402.6 for Ambrisentan respectively.

\textbf{Chromatography}

An analytical column of Waters Company, X-Bridge C18 (150 x 4.6 mm, 3.5\ensuremath{\mu}m) was used in this analysis.  Buffer was prepared with 2.5 g of hexane sulfonic acid in 1000mL water was adjusted to pH 2.5 by using orthophosphoric acid and mobile phase consists of buffer and acetonitrile in the ratio of 80:20 v/v, respectively. The source flow rate was 300 \ensuremath{\mu}l/min without split and the injection volume was 10 \ensuremath{\mu}l. Selexipag was eluted at 2.92 \ensuremath{\pm} 0.2 min and total injection run time of 10 min.Selexipag and Ambrisentan were eluted at retention time of 2.9 min and 7.4 min respectively.

\textbf{Linearity graph for standards and QC samples}

Two types of stock solutions of Selexipag were prepared using acetonitrile as solvent. One stock solution was used for calibration standards in method validation and the second stock solution was used for QC sample analysis. By using first stock solution calibration curve was drawn at different concentrations of 1, 2.5, 5, 7.5, 10, 12.5, 15 and 20 ng/ml by spiking suitable quantity of Selexipag in blank rat plasma. Similarly, quality control samples at various concentrations 1, 5, 10 and 15 ng/ml of Selexipag were prepared using second stock solution.

\textbf{Extraction of plasma}

Rats were placed in metabolic cages for urine collection after for 2 hours and stored in refrigerator. Rat plasma and Rabbit plasmas were collected by cardiac puncture, and from liver. To collect plasma, used tubes that contains dipotassium ethylene diamine tetra acetate (K\ensuremath{_{2}}EDTA) as an anticoagulant (used 10 \ensuremath{\mu}l of 0.1M K\ensuremath{_{2}}EDTA for 200-400 \ensuremath{\mu}l of plasma; ensured the concentration of K\ensuremath{_{2}}EDTA used does not interfere with the downstream assay) and stored in refrigerator. Human plasma was collected in tubes that contain K\ensuremath{_{2}}EDTA as an anticoagulant and stored in refrigerator.

\textbf{Preparation of Spiked Sample solution in Rat Plasma}

The test sample was prepared by mixing of 500\ensuremath{\mu}l of rat plasma, 500\ensuremath{\mu}l of acetonitrile, 500\ensuremath{\mu}l of internal standard (IS) and 500\ensuremath{\mu}l of the sample. The samples were precipitated from the respective solutions using vortex cyclo mixture for 10 minutes. Centrifuged at 4000 rpm for 20 mins. Later collected the supernatant solution in HPLC vial and injected into the chromatographic system.

\textbf{Preparation of Spiked Sample solution in Rabbit Plasma}

The test sample was prepared by mixing of 500\ensuremath{\mu}l of rabbit plasma, 500\ensuremath{\mu}l of acetonitrile, 500\ensuremath{\mu}l of internal standard (IS) and 500\ensuremath{\mu}l of the sample. The samples were precipitated from the respective solutions using vortex cyclo mixture for 10 minutes. Centrifuged at 4000 rpm for 20 mins. Later collected the supernatant solution in HPLC vial and injected into the chromatographic system.

\textbf{Preparation of Spiked Sample solution in Human Plasma}

The test sample was prepared by mixing of 500\ensuremath{\mu}l of human plasma, 500\ensuremath{\mu}l of acetonitrile, 500\ensuremath{\mu}l of internal standard (IS) and 500\ensuremath{\mu}l of the sample. The samples were precipitated from the respective solutions using vortex cyclo mixture for 10 minutes. Centrifuged at 4000 rpm for 20 mins. Later collected the supernatant solution in HPLC vial and injected into the chromatographic system.

\textbf{Preparation of Spiked Sample solution in Rat Urine}

The test sample was prepared by mixing of 500\ensuremath{\mu}l of human plasma, 500\ensuremath{\mu}l of acetonitrile, 500\ensuremath{\mu}l of internal standard (IS) and 500\ensuremath{\mu}l of the sample. The samples were precipitated from the respective solutions using vortex cyclo mixture for 10 minutes. Centrifuged at 4000 rpm for 20 mins. Later collected the supernatant solution in HPLC vial and injected into the chromatographic system.

\textbf{Selectivity}

In plasma selectivity measurements six different lots in rat plasma were taken and analyzed in three different concentrations such as high, medium and low level of quality control samples. 

\textbf{Effect of Matrix}

Effect of the matrix in Selexipag along with IS was evaluated by relating the area of the peak ratio of the extracted plasma sample which is taken from six different rats. The tests were carried out at middle quality control (MQC) levels in three replicates with six different plasma lots within the acceptable range i.e., \% CV \ensuremath{\leq } 15 \%.

\textbf{Accuracy and Precision}

It was obtained by the analysis of six QC samples at LLOQ QC, LQC, MQC and HQC levels. The acceptance criteria of \% CV should be below 15 \% for precision and accuracy should be within 15 \% except for LLOQ QC, which has 20 \% sample accuracy.

\textbf{Recovery}

The effectiveness of extraction of Selexipag was determined by analysis of each quality control sample for six replicates. By relating the peak areas of extracted standards and non-extracted standards percentage recoveries were estimated for Selexipag.

\textbf{Stability}

Stability analysis was performed by relating the peak area ratio for internal standard in stability sample and freshly prepared sample solution. Stability analysis was carried out for six preparations of LQC and HQC concentration levels. As per US FDA guideline \unskip~\citep{739766:17569147,739766:17569127} samples were stable below 15 \%. Benchtop stability samples and 2-8\ensuremath{^\circ}C in autosampler tested up to 24 hrs. Freeze-thaw stability samples were frozen at -30\ensuremath{^\circ}C and thawed three times compared to a freshly prepared quality control sample. Six different samples of HQC and LQC concentration levels were used for freeze-thaw stability. Later they were stored at below 10\ensuremath{^\circ}C to wet extracted stability samples for 24 hrs and compared the stability samples data with initially prepared fresh solutions data to assess the long term stability.
    
\section{Results and Discussion}
\textbf{Method Development}

In the method development process, different types of techniques were optimized in the mass spectrometry such as chromatography and extraction of the samples. Different trails were performed as shown in Table~\ref{tw-f59b203916f2} and their related figures are shown in Figures~\ref{f-b6caea5a6b6f}, \ref{f-61092186cb0c}, \ref{f-63089a65903c}, \ref{f-c9e5e9b9e9fc} and~\ref{f-ae40b2084f03}.

\textbf{Optimization of Mass spectrometry detection parameters}

The high response electrospray ionization (ESI)was selected for this study. The optimization of the instrument to give signal stability and sensitivity during infusion of the sample analyte in the prolonged flow rate of mobile phase to electrospray ion source maintained at both polarities. Selexipag give a high response in +ve ion mode when compared to {\textendash}ve ion mode. The major peaks in the primary ESI spectra of Selexipag of MH\ensuremath{^{+}}ions observed at m/z 462.5 and 496.2, respectively (Figure~\ref{f-d82d864b5527}). Internal standard Ambrisentan MH\ensuremath{^{+}} ions were observed at 378.5 and 402.6, respectively (Figure~\ref{f-fd994b7b7d94}).

\textbf{Chromatography Optimization}

Different types of mobile phases were used to develop the present methods presented in Table~\ref{tw-f59b203916f2}. The pKa value of Selexipag was found to be 3.77. In acidic buffers, sharp peaks were obtained. In all the trails only acidic buffers were used for method development. Plate count and tailing factor for Selexipag were not satisfactory with the mobile phase composition of Buffer (0.1 \% triethylamine pH adjusted to 2.5 with orthophosphoric acid) and acetonitrile in the ratio of 20:80 v/v. After that a mobile phase combination of hexane sulfonic acid/acetonitrile with the change in the composition was utilized for method development. Finally, the mobile phase composition of buffer (2.5 g hexane sulfonic acid in 1000 mL water with pH adjusted to pH 2.5 with orthophosphoric acid) and acetonitrile in the ratio of 80:20 v/v in gradient program was optimized. Later, an improvement in peak shape and peak response for Selexipag and internal standard was observed using the finalized mobile phase composition. An analytical column X-Bridge C18 (150 x 4.6 mm, 3.5 \ensuremath{\mu}m), Waters Company was finalized with a flow of 1.0 mL/min was used. The main analyte and internal standard were eluted within 10 min of run time (Table~\ref{tw-f59b203916f2}).

\textbf{Optimization of Sample Extraction}

Sample was extracted for removal of proteins from the prepared solution. At first tested with various extraction procedures like protein precipitation (PPT),  liquid- liquid extraction (LLE) and solid phase extraction (SPE). In solid phase extraction (Figure~\ref{f-9d61fde07332}) stationary phase made up of packed syringe-shaped cartridge, with 96 well plate and 47 mm flat disk packed with sorbent material in liquid handling syringe. Suppression effect was observed in protein precipitation method for internal standard and drug. In solid phase extraction, low analyte recovery was observed (80.56\%). In all the extraction procedures Precipitation extraction is suitable for extraction of the internal standard and Selexipag drug. Various organic solvents such as ethyl acetate, acetonitrile, chloroform, n-hexane, dichloromethane and methyl tertiary butyl ether individually or as well as with combination in PPT were used to extract the analyte from plasma sample. Among them acetonitrile produces appreciable extraction results by using Ambrisentan as internal standard for this analysis. Moreover, there is no significant effect in internal standard on analyte recovery, sensitivity or ion suppression. However, in Protein Precipitation method (Figure~\ref{f-401ed6a649df}), shows prominent recovery (92.64 \%) and selectivity was observed. 

Also observed appreciable recovery was observed while extracted in Rat plasma compared to Rabbit plasma, Human plasma and Rat urine in protein precipitation method.

These optimized parameters of the extracted procedure and chromatographic conditions bring about in reduced analysis time with accurate and precise detection of Selexipag in rat plasma.

\textbf{Method validation}

Validation of the analytical method for Selexipag assay in rat plasma was performed by as per ICH guidelines. This method was validated for the effect of the sample matrix, sensitivity, selectivity, accuracy, linearity, preciseness, reproducibility, and stability with high recoveries. 

\textbf{Sensitivity and Selectivity}

Blank and spiked sample with a lower limit of quantification quality control (LLOQ QC) in rat plasma for Selexipag are shown in Figure~\ref{f-de2e98ae52f5} and Figure~\ref{f-263f6722af10}. The \% interference of retention time of Selexipag peak was found to be 0.0\% among six various batches of rat plasma, comprising hemolyzed and lipidemic plasma containing K\ensuremath{_{2}}EDTA as an anticoagulant, and it is in acceptance criteria. Six replicates of Selexipag extracted samples at the LLOQ QC level in one of the plasma sample having the least interference at the Selexipag retention time were analyzed. The \% coefficient of variance for the peak isthe response for six sample replicate was found to be 1.2 \% for Selexipag. Similarly, blank and spiked sample with a lower limit of quantification quality control (LLOQ QC) in rabbit plasma, human plasma and rat urine for Selexipag are shown in Figures~\ref{f-c00fedce4e01}, \ref{f-9e9ee4dd42b6}, \ref{f-9c19dfe6177e}, \ref{f-6d6c7bd37f26}, \ref{f-dd304ba25634} and~\ref{f-1de126a74b9f}.

\textbf{Effect of Matrix}

For ion suppression/enhancement the \% CV for the signal response was observed 1.0 \% for MQC level for Selexipag, resembles that the effect of matrix on the ionization of analyte is well within the acceptable range under optimized conditions (Table~\ref{tw-711742c4376d}).

\textbf{Linearity}

The Selexipag area ratio was proportional to the concentration of Selexipag prepared and injected for calibration standards. The Selexipag assay was determined over the concentration range of 1 to 20 ng/mL of Selexipag analyte. The calibration curves show linearity and represented in Figure~\ref{f-353c714de4d4}. The correlation coefficient was \ensuremath{\geq } 0.999 (Table~\ref{tw-95998a124418}) for various concentration levels of Selexipag.

\textbf{Accuracy and Precision}

Accuracy and precision were evaluated by mixing all separate assay results obtained of six replicates in four different days for quality control samples. The \% CV for precision was \textless\ 5 \% and accuracy was obtained in a range of 98 to 101 for Selexipag in all the days (inter-run). All the outcome of this data tabulated in Table~\ref{tw-c8dfd0913546} and shows the method accuracy and precision.

\textbf{Recovery}

Six aqueous spiked sample solutions at three different concentrations low, medium an high-quality control levels for Selexipag were prepared for recovery estimation, and the peak areas occurred for extracted samples of the same strength level from a precision and accuracy samples on the same day. The average recovery for Selexipag was 98.2 \% with a precision of 1.2 \%. This specifies that the extraction efficiency for Selexipag was reliable and reproducible. 

Also found reproducible recoveries were observed for Selexipag in rat plasma while in comparison with rabbit plasma, human plasma and rat urine (Table~\ref{tw-891b4369db50}). 

\textbf{Reinjection Reproducibility}

This test was carried to verify whether the instrument performance remains unaffected due to any instrument failure during sample analysis. The change was less than 2.0 at LQC and HQC concentration levels at initial and even after analyzed 24 hr shows less \% change and hence sample can analysis in case of instrument breakdown. 

\textbf{Stability Studies}

In solution stability analysis, Selexipag in solutions were prepared in diluent and put in storage at 2-8\ensuremath{^\circ}C in a refrigerator. Fresh stock solutions were related to aged stock solutions prepared earlier 24hr. The \% change for Selexipag was 1.04 \%, which shows that stock solutions were stable at least 24hr when stored in 2-8\ensuremath{^\circ}C. Selexipag samples in bench top and instrument autosamplers were analyzed at HQC and LQC levels. Selexipag was stable in plasma for at least 24hr when stored at room temperature and 24hr in an autosampler when stored at 20\ensuremath{^\circ}C. Hence it indicates that repeated freezing and thawing (three cycles) of samples in plasma spiked with Selexipag at HQC and LQC levels doesn't impact their solution stability. The stability analysis results are tabulated below (Table~\ref{tw-2906e72760bc}) specifies that Selexipag was stable in a matrix up to 24hr at a storage condition of -30\ensuremath{^\circ}C.

\textbf{Application to pharmacokinetic study}

The validated method has been successfully quantified the Selexipag concentration in various three groups of rats, under fasting conditions after administration of 800 mcg Selexipag drug sample was injected into a rat body collected the samples at different time intervals like 0.5, 1.0, 1.5, 2.0, 2.5 and 3.0 hours. After that samples were prepared as specified method and injected in the chromatographic system and recorded the results. The pharmacokinetics parameters assessed were C\ensuremath{_{max}} (maximum observed drug concentration during the study). The values of AUC\ensuremath{_{0-12}} (area under the plasma concentration with time curve measured 2.0hr, using the trapezoidal rule), T\ensuremath{_{max}} (time to observed maximum drug concentration), K\ensuremath{_{el}} (apparent first order terminal rate constant calculated from a semi-logplot of the plasma concentration versus time curve, using the method of the least square regression) and t\ensuremath{_{1/2\ }} (terminal half-life as determined by the quotient 0.693/K\ensuremath{_{el}}) are tabulated in Table~\ref{tw-3b347697bda3}.

The test/reference ratios for C\ensuremath{_{max}}, AUC\ensuremath{_{0-12}}, and AUC were 80.56, 90.12 respectively, and they were within the acceptance range of 80\%-125\%demonstrating the bio-equivalence of the formulation of Selexipag (Figure~\ref{f-d93420f77a65}). 
    
\section{Conclusions}
The present proposed developed and validated quantification method shows high sensitive LC-ESI-MS/MS method for the estimation of Selexipag in rat plasma. Moreover, this method is fast, rugged, reproducible bio analytical method to be developed LC-ESI-MS/MS method for the estimation of Selexipag in rat plasma. The method was developed to assess the lower limit of detection (0.5 ng/mL) and limit of quantification (5 ng/mL) concentrations in rat plasma. Linearity was observed in calibration curves for assay Selexipag in a range of 1 ng/mL to 20 ng/mL of concentration in rat plasma for each analyte and its regression correlation coefficient was found to be 0.999. This analytical method is simple and well-organized and can be utilized in pharmacokinetics studies and also for monitoring of samples injected in body fluids too. We strongly believe that, this study can provide a road map for developing various new quantification methods for various drug molecules. 

\textbf{Conflicts of interest}

The authors declare that there are no conflicts of interest.

\textbf{Ethical department details for animal studies}

Registration No. Nirmala college of pharmacy, 1629/PO/a/12/CPCSEA.
\section*{Acknowledgement}N. L. A. Amara babu thanks VFSTR for the research avenues and facilities.  Dr. K. P. Rao Thanks DST-SERB, for financial support under project for early career, project no. EMR/2014/001114.

\textbf{Abbreviations}

LOQ: Low Quality control; HQC: High Quality control; MQC: Medium Quality Control; PPE: Protein precipitation extraction; LLE: Liquid liquid extraction; T\ensuremath{_{max}}: Time required for C\ensuremath{_{max}}; Min: Minutes; ng/mL: Nano gram per milli liter; RSD: Relative standard deviation; ICH: International council for harmonization; LOQ: Limit of quantitation; LOD: Limit of detection; LLQC: Low level Quality control.
    

\bibliographystyle{pharmascope_apa-custom}

\bibliography{\jobname}
\end{document}
